Homework 7 CS420/KAIST Spring 1999

ff aŰ

Due: May 12 (¡Åffi), 24:00

1999[ 5Ú 6ffi

řü˚ Exerciseffiù [¸Ć ȷĄľt SpĆ3 (SEô 3221/3222) C;ć ř. Exercise 1 \Lemma Proof" (30pts)

Complete the proof of Lemma 2 at page 10 of the lecture notes. 2

Exercise 2 \Soundness Proof" (30pts)

Complete the proofs for the tcoML type’s strong soundness (Theorem 2 at page 9 of the lecture notes). 2

Exercise 3 \Incomplete Examples" (10pts)

tcoML˚ Í«èSSðù K;ăŞ� ř;ăŞŹ Mř. Ň, [C{¸ îI`Ź tcoML Š£ŤŰ¸ Í«*Š¸ üŞ MŹ EÍ` –ř. Ťăć Č`Ş Žõ ffiĘ. 2